Author's response to reviews

Title: The role of CACNA1S in predisposition to malignant hyperthermia

Authors:

Danielle Carpenter (danielle.carpenter@nottingham.ac.uk)
Christopher Ringrose (chrisringrose81@hotmail.com)
Vincenzo Leo (vincenzoleo1@hotmail.com)
Andrew Morris (amorris@well.ox.ac.uk)
Rachel L Robinson (r.l.robinson@leeds.ac.uk)
P Jane Halsall (p.j.halsall@leeds.ac.uk)
Philip M Hopkins (p.m.hopkins@leeds.ac.uk)
Marie-Anne Shaw (m.shaw@leeds.ac.uk)

Version: 5 Date: 2 October 2009

Author's response to reviews: see over
Dear Editors

Please find an amended manuscript to accommodate the changes requested

Formatting changes requested
---------------------------------
Major revisions (We require the author to make these changes)

References - we cannot currently find references 46 cited in the manuscript text. Please ensure that all the references are cited in the text in numerical order. Any references not cited in the text will not be included in the final publication.

The references have been checked and reference 46 has now been included in table 2, from where it was accidentally omitted.

Thank you

Danielle carpenter